Author's response to reviews

Title: Fertility in four regions spanning large contrasts in serum levels of widespread persistent organochlorines: a cross-sectional study

Authors:

Gunnar Toft (gutof@as.aaa.dk)
Anna Axmon (anna.axmon@med.lu.se)
Aleksander Giwercman (aleksander.giwercman@kir.mas.lu.se)
AneMarie Thulstrup (anemt@as.aaa.dk)
Anna Rignell-Hydobom (anna.rignell-hydobom@ymed.lu.se)
Henning Sloth Pedersen (hsp@natur.gl)
Jan Ludwicki (K.LUDWICKI@PZH.GOV.PL)
Valentina Zvyesday (dimusic@ic.kharkov.ua)
Alex Zhinchuk (dimusic@ic.kharkov.ua)
Marcello Spano (spanomrc@casaccia.enea.it)
GianCarlo Manicardi (manigi04@mail.unimo.it)
Eva Bonefeld-Jorgensen (ebj@mil.au.dk)
Lars Hagmar (lars.hagmar@ymed.lu.se)
Jens Peter Bonde (jpbon@as.aaa.dk)

Version: 5 Date: 26 October 2005

Author's response to reviews:

Dear Editors:

We are delighted to learn that the paper has been accepted for publication in BMC EH and look forward to see our work appearing in this distinguished journal.

The uploaded third version of the manuscript has been revised to fulfil all requests with respect to the format. Thus we have responded to each of the mentioned enquiries:

1. Email addresses on title page and address for Inuendo corrected
2. Country names in the Abstract
3. Line and section spacing
4. No dots in headings
5. Reformatting of tables to allow portrait format
6. All authors mentioned in reference list (no 'et al') and several minor corrections.

Moreover, we have been reading all the instructions carefully once again, reviewed the WORD template and hope that the manuscript is now exactly matching the format instructions,

yours sincerely, on behalf of all authors,

Jens Peter Bonde